Author's response to reviews

Title: fish bone foreign body presenting with acute fulminating retropharyngeal abscess in a resource challenged center: case report

Authors:

   Olushola A Afolabi (droaafolabi@yahoo.com)
   Joseph O Fadare (jofadare@gmail.com)
   Ezekiel O Oyewole (droyewole2002@yahoo.com)
   Stephen A Ogah (stephenogah@yahoo.com)

Version: 2 Date: 27 August 2010

Author's response to reviews: see over
The editor,

Journal of medical Case report

Dear Sir

COVERING LETTER FOR THE MANUSCRIPT TITLED: FISH BONE FOREIGN BODY PRESENTING WITH ACUTE FULMINATING RETROPHARYGEAL ABSCESS IN A RESOURCE CHALLENGE CENTER: CASE REPORT

The authors wish to submit inform the editor that the correction noted has been effected.

The title page now has the title, the full names of authors, email addresses and institutional address.

The second page now contains the abstract in the structured format of introduction, case presentation and conclusion, so also is the main article based on the instruction to authors.

The referencing style has also been corrected corresponding to the journal style.

The figures have now been made separately in different files.

Thanking you in anticipation

Yours Faithfully,

Signed
Dr O.A. Afolabi

(Correspondence author)